 Ev/23-Jun-05
Week 25  starting 20 June 05
SM18 Operation Meeting (Tuesdays)

Bench status  on Monday  20 June 05 @: 20/06/2005 07:53:29
	LHC SM18 Magnet Test Facility 


	11
	A1
	MBAR3343
	PT 14 Magnet MAPS
	299.92

	10
	A2
	MBBL1235
	ICS 4 Final connection
	298.94

	3
	B1
	SSS223
	PT 1 IAP @ cold
	1.90

	9
	B2
	SSS219
	Prep 5 Cool Down
	73.55

	4
	C1
	MBAR3252
	Prep 5 Cool Down
	4.66

	2
	C2
	MBAL2191
	PT 6.8 Training 8
	1.90

	12
	D1
	SSS200
	PT 13.2 Resist. Meas.
	297.41

	1
	D2
	SSS616
	Prep 5 Cool Down
	205.28

	6
	E1
	MBAL3103
	PT 12 Warm Up
	135.88

	7
	E2
	MBBR3362
	Prep 5 Cool Down
	84.54

	8
	F1
	MBBR3361
	PT 12 Warm Up
	206.66

	5
	F2
	MBAL2196
	Prep 5 Cool Down
	94.06


   Temperature data are taken from TBSM18 site
Coordinator this week: S. Sanfilippo;
MTM support on call : R. Mompo  16-3602 
16  complete Magnet Tests cycles   ( S277  ROBTC )
10  dipoles( 2 with MM )+ 5  SSS( 0  with MM) +  1 SSSS (with MM)
[SSSS639 put in standby & 1236 put in standby because Cryo Leak]

   Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight) 
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
· General Points 
· TV Network maintenance (Should inform SMCR next time 15-06-05).
· Vtaps or current lead Wrongly connected by ICS (To enforce again connection quality)

· Pumping down difficult for Cryo (Saturday problem).

· 1236 leak inside magnet.
· Mobile phone lost by ICS.
· AOB

Tests Information 

	Magnets under Test from 12-06-2005 to 18-06-2005 

	-
	Bench
	Magnet
	Shafts
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for

	1
	TBA1
	MBAR3305 
	[image: image1.png]



	08-06-2005 
	23 h.
	59 h.
	21 h.
	14 h.
	15 h.
	14-06-2005 
	132 h.
	Stripping

	2
	TBA2
	MBAR3352 
	[image: image2.png]



	08-06-2005 
	24 h.
	28 h.
	39 h.
	18 h.
	8 h.
	13-06-2005 
	117 h.
	Stripping

	3
	TBB1
	SSS283 
	[image: image3.png]



	14-06-2005 
	24 h.
	13 h.
	20 h.
	7 h.
	15 h.
	17-06-2005 
	79 h.
	Stripping

	4
	TBB1
	SSS285 
	[image: image4.png]



	10-06-2005 
	29 h.
	12 h.
	16 h.
	6 h.
	14 h.
	13-06-2005 
	77 h.
	Stripping

	5
	TBB2
	SSS213 
	[image: image5.png]



	13-06-2005 
	32 h.
	13 h.
	18 h.
	12 h.
	16 h.
	17-06-2005 
	91 h.
	Stripping

	6
	TBB2
	SSS220 
	[image: image6.png]



	09-06-2005 
	29 h.
	14 h.
	15 h.
	14 h.
	6 h.
	12-06-2005 
	78 h.
	Stripping

	7
	TBC1
	MBAL1224 
	[image: image7.png]



	13-06-2005 
	26 h.
	22 h.
	18 h.
	13 h.
	21 h.
	17-06-2005 
	100 h.
	Stripping

	8
	TBC1
	MBAR3277 
	[image: image8.png]



	09-06-2005 
	22 h.
	22 h.
	15 h.
	15 h.
	8 h.
	13-06-2005 
	82 h.
	Stripping

	9
	TBC2
	MBBL3100 
	[image: image9.png]



	11-06-2005 
	31 h.
	23 h.
	15 h.
	14 h.
	10 h.
	15-06-2005 
	93 h.
	Stripping

	10
	TBD1
	SSS255 
	[image: image10.png]



	11-06-2005 
	35 h.
	31 h.
	28 h.
	15 h.
	27 h.
	16-06-2005 
	136 h.
	Stripping

	11
	TBD2
	SSS639 
	[image: image11.png]



	07-06-2005 
	101 h.
	21 h.
	59 h.
	11 h.
	31 h.
	18-06-2005 
	223 h.
	Stand By

	12
	TBE1
	MBBL1231 
	[image: image12.png]



	11-06-2005 
	38 h.
	21 h.
	17 h.
	18 h.
	8 h.
	16-06-2005 
	102 h.
	Stripping

	13
	TBE2
	MBBL1236 
	[image: image13.png]



	10-06-2005 
	33 h.
	0 h.
	0 h.
	1 h.
	6 h.
	12-06-2005 
	40 h.
	Stand By

	14
	TBE2
	MBBL3125 
	[image: image14.png]



	12-06-2005 
	26 h.
	23 h.
	22 h.
	13 h.
	17 h.
	16-06-2005 
	101 h.
	Stripping

	15
	TBF1
	MBAR3347 
	[image: image15.png]



	09-06-2005 
	28 h.
	95 h.
	17 h.
	17 h.
	11 h.
	16-06-2005 
	168 h.
	Stripping

	16
	TBF2
	MBAL1225 
	[image: image16.png]



	07-06-2005 
	26 h.
	32 h.
	157 h.
	13 h.
	25 h.
	18-06-2005 
	253 h.
	Stripping


Quench Information
	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10

	1
	MBAR3305
	12385 A.
(D2-U)
	12501 A.
(D2-L)
	12710.9 A.
(D2_U)
	[image: image17.png]



	-
	-
	-
	-
	-
	-

	2
	MBAR3352
	12488 A.
(D1-L)
	0 A.
(No Data)
	12851 A.
[image: image18.jpg]



	-
	-
	-
	-
	-
	-
	-

	3
	SSS283
	11974 A.
(QE_12)
	12141 A.
(QI_12)
	12220 A.
(QI_34)
	[image: image19.png]



	-
	-
	-
	-
	-
	-

	4
	SSS285
	11835 A.
(QE-12)
	12342.2 A.
(QE_34)
	12442.9 A.
(QE-34)
	[image: image20.png]



	-
	-
	-
	-
	-
	-

	5
	SSS213
	12833 A.
(QI_12)
	12850 A.
[image: image21.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	6
	SSS220
	11250 A.
(QI-12)
	12358 A.
(QE-12)
	12539 A.
(QE12)
	[image: image22.png]



	-
	-
	-
	-
	-
	-

	7
	MBAL1224
	10845 A.
(D2-L)
	12512 A.
(D1-L)
	12620 A.
(D1-L)
	[image: image23.png]



	-
	-
	-
	-
	-
	-

	8
	MBAR3277
	11371 A.
(D2-U)
	12394.6 A.
(D2-U)
	12506 A.
(D1-U)
	[image: image24.png]



	-
	-
	-
	-
	-
	-

	9
	MBBL3100
	11527 A.
(D1_U)
	12850 A.
(D2_L)
	12850 A.
[image: image25.jpg]



	-
	-
	-
	-
	-
	-
	-

	10
	SSS255
	11260.1 A.
(QI_12)
	12168 A.
(QI_12)
	12356 A.
(QI_34)
	[image: image26.png]



	-
	-
	-
	-
	-
	-

	11
	SSS639
	5820 A.
[image: image27.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBBL1231
	12050 A.
(D2-U)
	12525 A.
(D1-U)
	12857 A.
(D2-U)
	[image: image28.png]



	-
	-
	-
	-
	-
	-

	13
	MBBL1236
	[image: image29.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	14
	MBBL3125
	11858.1 A.
(D2_L)
	12680.1 A.
(D2_U)
	12850 A.
[image: image30.jpg]



	-
	-
	-
	-
	-
	-
	-

	15
	MBAR3347
	12719.9 A.
(D1-U)
	9860 A.
(D1-U)
	12850 A.
[image: image31.jpg]



	-
	-
	-
	-
	-
	-
	-

	16
	MBAL1225
	11347 A.
(D1-L-S01)
	12744 A.
(No Data)
	12850 A.
[image: image32.jpg]



	-
	-
	-
	-
	-
	-
	-



[image: image33.png]o
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	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday _14_A
	TBD1


	 SSS255: Dummy resistor blown up when discharging quench heater capacitor
	MTM
	
	Salem replaced the dummy resistor.

	Tuesday _14_A
	TBE2


	 No data in IAP file
	MTM
	
	Salem found that there was no trigger hf.  Test repeated after “ reset”.

	Tuesday _14_A
	TBD1


	 SSS255: Not able to power CD circuit

During IAP at warm : error confirmed between the two circuits (MQT,sextupole)
	ICS
	
	Investigation by Salem.

CD and RF circuit inverted during the connection (CD leads to EF bus-bar & EF leads to CD bus-bar)

	Tuesday _14_N
	TBA2


	 Cryo leak in MB2524 (MRB side).
	ICS
	
	Joint forgotten.

	Wednesday _15_M
	TBF2


	SPA during machine cycle on MB1225.
	ACR?

AB/CO?


	
	Temperature auxiliary fault ?

Communication problem?



	Wednesday _15_A
	TBD2


	SPA during SSW measurements.
	?
	
	Monitoring of SPA will be installed by Giorgio in the PLC.

	Wednesday _15_N
	TBD2


	MBBL3125: Resistance measurements in CDW2 show that this magnet has a diode.

	?
	
	Information provided in the file http://lhcdipcoor.web.cern.ch/LHC-dipcoor/diodes/dioderesume.pdf for magnet MBBL3125 is wrong.

Diode test was not performed.
The diode test has to be performed in any case.

	Wednesday _15_N
	TBF2


	Machine cycle LHC_Pf_1800 (last test) was stopped during the measurement plateau because of a SPA
	AB/PO?

MTM


	
	N.Sammut (contact person) was not informed. The test was not repeated. 

NB: This test is dedicated to check the influence of a long pre_injection porch and the data coming from it are very important

	Thursday _16_M
	TBF2


	The SSL quench at 4K is with a current of 10173A at 1350mBar higher than tolerance (<10100A),
	MTM


	
	Salem contacted. Decision was to warm up the magnet.

	Thursday _16_A
	TBE1


	Permanent Triggers present on VSupra_ext & VClamp_Ext for preparatory test PT2.1.
	MTM


	
	Salem has by-passed the Vtaps. Ok to continue.

	Thursday _16_A
	TBE1


	MBAL 3103, Prep2.1: the value of  Quench Heater H4,TminV (=1741ms), is too high.


	MTM


	
	Salem checked the file and gave the OK to continue because the problem is coming from the QH power supply.

This power supply was changed.

	Thursday _16_N


	TBA2


	MB2524: Data from AQA not available after quench. Lf config file not found in control room sun. Error displayed in local sun : Error 11 occurred at open file

	AB/CO


	lf
	Data could not be validated in AQA immediately. But quench analysed locally. Then Salem transferred the data to the control room sun.



	Friday_17_M
	TBA2


	RMA cannot be restarted.
	AB/PO


	
	AB-PO experts called by Salem. Test in short circuit. Firing card changed.

	Friday _17_N
	TBA1


	AC offset for D1_L-U is very high. Equipment support will look after this problem.

D1-D2 signal was flat. Equipment support has been contacted and Salem told that he will look for the problem and inform


	MTM


	
	Salem changed the potaim card

Salem changed the potaim card.

	Saturday _18_M
	
	One analyser pumping unit is not working.

Cooling down slow.
	ACR


	
	J.P.Lamboy gave some instruction.



	Saturday _18_M
	
	General stop due to water cut. Liquefactor stopped.. Magnets are warming up.
	?


	
	Stop of all the tests at 1.9 K up to the afternoon



	Saturday _18_M
	TBB2
	IAP@warm 2.2C (corrector clamp check) on SSS219 could not be launched, triggers present on V_clamp E-F.
	ICS


	
	ISC team was asked to check the connections.Clamp re-done

	Sunday _19_A
	TBD2


	ICS found that one heating circuit of the anticryostat on SSS616 is opened.


	MTM


	
	Salem came and checked with ICS: no open circuit.

But he found that the anticryostat parameters were wrong because of a new anticryostat. New parameters implemented by P.Legrand.

	Sunday _19_A

Monday_20_M
	TBC2

TBD2


	Liquefactor problem: BP too high. Regulation difficult. Low capacity of pumping.

Cooling of special SSS slow: Cryo team encountered several problems during the night causing very slow recovery rates on all benches. 

Another problem with BP too high


	ACR


	
	

	Sunday _19_N
	TBD1


	SSS200: during resistance measurements (PT 13.2): Corrector Voltage Taps (E & F) and Sextupole Voltage Taps (C & D) seems to be inverted.
	MTM


	
	Salem advised to check with ICS during the magnet opening.

SSS200 had a new configuration of connection: the values are OK.

	Sunday _19_N
	TBC2


	Difference in current reading between AQA and Voltmeter (more than 10 Amps)


	MTM


	AQA?
	Checked Monday for the next quenches. The ADC card may be changed.

	Monday _20_A
	TBA1


	HV test could not be started.

Voltage exceeded the upper limit.
	MTM


	
	Investigation by Richard.











